Author’s response to reviews

Title: “Posting policies don’t change because there is peace or war”: staff deployment challenges for two large health employers during and after conflict in Northern Uganda

Authors:

Richard Mangwi Ayiasi (mangwirichard@gmail.com)
Elizeus Rutebemberwa (ellie@musph.ac.ug)
Tim Martineau (Tim.Martineau@lstmed.ac.uk)

Version: 1 Date: 11 Feb 2019

Author’s response to reviews:

We are grateful for the comments that we received from the two reviewers. These comments have made our paper even stronger. We have taken time to respond/address to each and every comment made. This has been presented in a tabulated manner with two columns. The first column has the reviewers' comments and the second column has the authors responses. Please note that line numbers and reference numbers have changed due to the various adjustments and rearrangements. However, we provide in each response, a reference to where in the text, such adjustments have been made. Also attached is a copy of the tracted changes of the manuscript for easy following.